Author's response to reviews

Title: Measuring patient safety culture in Taiwan: an application of Hospital Survey on Patient Safety Culture (HSOPSC) in Eastern culture

Authors:

I-Chi Chen (FL001@mail.oit.edu.tw)
Hung-Hui Li (hhlee@saturn.yzu.edu.tw)

Version: 2 Date: 6 October 2009

Author's response to reviews:

October 5, 2009

Editorial Board,
BMC Health Services Research

Dear Editors,

I am submitting a paper entitled “Measuring patient safety culture in Taiwan: an application of Hospital Survey on Patient Safety Culture (HSOPSC) in Eastern culture” for consideration of publication in the BMC Health Services Research. This paper describes the results of applying the Hospital Survey on Patient Safety Culture (HSOPSC) questionnaire in assessing the culture of patient safety in Taiwan and gives explanation of some of the phenomena in patient safety culture that are unique in Eastern culture. The main contribution of this work is that it provides health care organizations in Taiwan a different perception about patient safety culture, and it explores the differences in patient safety culture between the West and the East.

Thank you. I look forward to hearing from you soon.

Sincerely yours,

I-Chi Chen
Lecturer
Department of Health Care Administration,
Oriental of Technology,
58, Sec 2, Sihchuan Road, Pan Chiao City, Taipei County, Taiwan, R.O.C.
Tel: +886-2-77380145 ext 6223
Fax: +886-2-77380898
Email: FL001@mail.oit.edu.tw